Author’s response to reviews

Title: The severe hypertension detection in patient with neurofibromatosis type 1 during anesthesia induction: A case report

Authors:

Juan Wang (534606134@qq.com)
Guohua Wei (18262639792@163.com)
Zhongyun Wang (791960701@qq.com)
Huang He (353550575@qq.com)

Version: 2 Date: 10 Oct 2019

Author’s response to reviews:

Thanks for the significant comments. This kindly advice is very helpful. We have made a careful revise, please kindly see the Figure 1.